Author's response to reviews

Title: Identifying context factors explaining physician's communication behaviour: an explorative study in general practice

Authors:

Geurt TJM Essers (g.essers@elg.umcn.nl)
Sandra M van Dulmen (S.vanDulmen@nivel.nl)
Chris van Weel (C.vanWeel@elg.umcn.nl)
Cees van der Vleuten (C.vanderVleuten@maastrichtuniversity.nl)
Anneke Kramer (a.kramer@elg.umcn.nl)

Version: 2 Date: 19 May 2011

Author's response to reviews:

Informed consent given by GPs and patients was explicitly added in the revised version of the article.